Author’s response to reviews

Title: Rationale and Design of the Brigham Cohort for Psoriasis and Psoriatic Arthritis Registry (COPPAR)

Authors:

Maria Schneeweiss (maria.schneeweiss@tufts.edu)
Joseph Merola (jfmerola@bwh.harvard.edu)
Elizabeth Karlson (ekarlson@partners.org)
Daniel Solomon (dsolomon@partners.org)

Version: 4 Date: 26 Jul 2017

Author’s response to reviews:

July 26, 2017

Editor-in-Chief

Professor Guangde Tu

BMC Dermatology

BDER –D-16-00054, R4

Rationale and Design of the Brigham Cohort for Psoriasis and Psoriatic Arthritis Registry (COPPAR)

Maria Schneeweiss; Joseph F. Merola; Elizabeth W. Karlson; Daniel H. Solomon

Dear Prof. Tu,

Thank you for your kind invitation to resubmit our article for your section on Study Protocols. We have now reformatted the article and abstract according to the guidelines for Study Protocol. All previous changes in response to reviewer comments are included as well.
The manuscript is not submitted or under review elsewhere. The funding sources and potential conflicts are disclosed at the appropriate places on your webpage and the manuscript.

Please do not hesitate to contact me for any questions.

Sincerely,

Maria Schneeweiss